%~MoulinÃ© par MaN_auto v.0.33.8 (9c3f6e85) 2025-05-09 15:28:43
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{\'Equations aux dÃ©rivÃ©es partielles}
\TopicEN{Partial differential equations}

\addbibresource{CRMATH_Le_20240596.bib}

\newcommand{\rad}{\mathrm{rad}}
\newcommand{\loc}{\mathrm{loc}}

\newcommand*{\dd}{\mathrm{d}}

\DeclareMathOperator{\divergence}{div}

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to macro
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}
%% Remplacer \xrightarrow{#1} par \overset{#1}{\longrightarrow}. 

%%% \mapsto macro
\let\oldmapsto\mapsto
\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}

%%% \longhookrightarrow macro
%\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}

%%% \longrightrightarrows macro
%\newcommand{\longrightrightarrows}{\overset{\raisebox{-.17ex}{$\displaystyle{\longrightarrow}$}}{\smash{\raisebox{-.3ex}{$\longrightarrow$}}}}

%%% Convergence
%\newcommand\dconverge[2][p \to + \infty]{\xrightarrow[#1]{#2}}
%\newcommand\tconverge[1][p \to + \infty]{\underset{#1}{\longrightarrow}}

%%% Isomorphism
%\newcommand{\isomto}{\overset{\!\!\raisebox{.3ex}{$\scriptstyle{\sim}$}}{\smash{\to}}}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

%%% VERTICAL BAR

%%% ''Evaluated at'' macro
\newcommand\restr[2]{{				% we make the whole thing an ordinary symbol
 \left.\kern-\nulldelimiterspace 	% automatically resize the bar with \right
 #1 									% the function
 \vphantom{\big\vert} 				% pretend it's a little taller at normal size
 \right\vert_{#2} 					% this is the delimiter
 }}
% As a subscript: \phi_{F_f\rvert_{K_\ell}} [NO : \phi_{\left.h\right\vert_{K_\ell}}].
% In textstyle : \bigr\rvert_{}.

%%% ''Such that'' macro
\newcommand*{\st}{\;\vert\;}
\newcommand*{\bigst}{\;\big\vert\;}
\newcommand*{\Bigst}{\;\Big\vert\;}
\newcommand*{\biggst}{\;\bigg\vert\;}
\newcommand*{\Biggst}{\;\Bigg\vert\;}
% Scaling version (with left and right delimiters): \,\middle\vert\,

%%% Divides
% Use \,\big\vert\,
% Or if \nmid is also present (?):
%\newcommand{\zerodel}{.\kern-\nulldelimiterspace}
%\newcommand{\divides}[2]{\left\zerodel#1\ \middle\vert\ #2\right\zerodel}

%%% Separator (non-scaling)
% \mid

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}\relabel}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}\relabel}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}\relabel}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=12.3cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{Liouville theorems for $p$-Laplace equations with critical Hardy--H\'enon exponents}
\alttitle{ThÃ©orÃ¨mes de Liouville pour les Ã©quations de $p$-Laplace avec des exposants critiques de Hardy--HÃ©non}

\author{\firstname{Diem Hang T.} \lastname{Le}}
\address{Faculty of Mathematics and Applications, Saigon University, 273 An Duong Vuong St., Ward 3, District~5, Ho Chi Minh City, Vietnam}
\address{Faculty of Data Science in Business, Ho Chi Minh University of Banking, Ho Chi Minh City, Vietnam}
\email{hangltd@hub.edu.vn}

\author{\firstname{Phuong} \lastname{Le}\IsCorresp\CDRorcid{0000-0003-4724-7118}}
\address{Faculty of Economic Mathematics, University of Economics and Law, Ho Chi Minh City, Vietnam}
\address{Vietnam National University, Ho Chi Minh City, Vietnam}
\email{phuongl@uel.edu.vn}

\subjclass{35J92, 35B53, 35B33, 35B38}
\keywords{\kwd{$p$-Laplace equations} \kwd{Hardy--H\'enon exponents} \kwd{Liouville theorems} \kwd{Palais--Smale sequences}}
\altkeywords{\kwd{\'Equations de $p$-Laplace} \kwd{exposants de Hardy--HÃ©non} \kwd{thÃ©orÃ¨mes de Liouville} \kwd{suites de Palais--Smale}}

\thanks{This research is funded by Vietnam National Foundation for Science and Technology Development (NAFOSTED) under grant number~101.02-2023.35}
\CDRGrant[Vietnam National Foundation for Science and Technology Development (NAFOSTED)]{101.02-2023.35}

\begin{abstract}
This paper is concerned with possibly sign-changing solutions to the critical $p$-Laplace equation $-\Delta_p u = \lvert{x}\rvert^\alpha \lvert{u}\rvert^{p^*_\alpha-2}u$ in $\Omega$, $u=0$ on $\partial\Omega$, where $1<p<N$, $\alpha>-p$, $p^*_\alpha=\frac{(N+\alpha)p}{N-p}$, and $\Omega$ is a bounded or unbounded domain of $\smash{\mathbb{R}^N}$. We first derive a Liouville type theorem on half-spaces. Then we classify solutions via the radial symmetry and Morse index. Moreover, we characterize the compactness of radial Palais--Smale sequences on radial domains.
\end{abstract}

\begin{altabstract}
Cet article s'intÃ©resse aux solutions de l'Ã©quation critique de p-Laplace qui peuvent changer de signe. $-\Delta_p u = \lvert{x}\rvert^\alpha \lvert{u}\rvert^{p^*_\alpha-2}u$ dans $\Omega$, $u=0$ sur $\partial\Omega$, oÃ¹ $1<p<N$, $\alpha>-p$, $p^*_\alpha=\frac{(N+\alpha)p}{N-p}$, et $\Omega$ est un domaine bornÃ© ou non de $\smash{\mathbb{R}^N}$. Nous dÃ©rivons d'abord un thÃ©orÃ¨me de type Liouville sur les demi-espaces. Ensuite, nous classons les solutions en fonction de la symÃ©trie radiale et de l'indice de Morse. De plus, nous caractÃ©risons la compacitÃ© des suites de Palais--Smale radiales sur les domaines radiaux.
\end{altabstract}

\begin{document}

\maketitle

\section{Introduction}

Unless specified, in this paper we always assume that $N\ge2$, $1<p<N$, $-p<\alpha<+\infty$ and $\Omega$ is a smooth possibly unbounded domain of $\mathbb{R}^N$. We are concerned with the critical Hardy--H\'enon problem
\begin{equation}\label{main}
\begin{cases}
-\Delta_p u = \abs{x}^\alpha \abs{u}^{p^*_\alpha-2}u	&\text{in $\Omega$},\\
u=0 											&\text{on $\partial\Omega$},
\end{cases}
\end{equation}
where $\Delta_p u ={\divergence}\parens[\big]{\abs{\nabla u}^{p-2}\nabla u}$ is the $p$-Laplacian and $p^*_\alpha = \frac{(N+\alpha)p}{N-p}$ is the critical Hardy--H\'enon--Sobolev exponent. To introduce a functional setting for~\eqref{main}, we recall the homogeneous Sobolev space
\[
\mathcal{D}^{1,p}(\mathbb{R}^N)=\braces[\bigg]{u\in L^{p^*}(\mathbb{R}^N):\nabla u\in L^p(\mathbb{R}^N;\mathbb{R}^N)},
\]
which is endowed with the norm
\[
\norm{u} = \parens[\Bigg]{\int_{\mathbb{R}^N} \abs{\nabla u}^p \dd x}^\frac{1}{p}.
\]
When $-p<\alpha\le0$, problem~\eqref{main} is related to Hardy--Sobolev's inequality
\begin{equation}\label{HSI}
S_\alpha \parens[\Bigg]{\int_{\mathbb{R}^N} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x}^\frac{p}{p^*_\alpha} \le \int_{\mathbb{R}^N} \abs{\nabla u}^p \dd x
\end{equation}
which holds for all $u \in \mathcal{D}^{1,p}(\mathbb{R}^N)$. The best Hardy--Sobolev constant $S_\alpha>0$ is defined as
\[
S_\alpha = \inf\braces[\Bigg]{ \int_{\mathbb{R}^N} \abs{\nabla u}^p \dd x : \int_{\mathbb{R}^N} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x=1 }
\]
(see~\cite{MR768824,MR1223899}). In the case $\alpha>0$, inequality~\eqref{HSI} holds for all $u \in \mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)$, where we denote by $\mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)$ the subspace of radial functions in $\mathcal{D}^{1,p}(\mathbb{R}^N)$, i.e.,
\[
\mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N) = \braces[\bigg]{u\in\mathcal{D}^{1,p}(\mathbb{R}^N) : u(x) = u(\abs{x})}
\]
(see~\cite{MR2422077}).

We denote by $\mathcal{D}^{1,p}_{0}(\Omega)$ the closure of $\mathcal{D}(\Omega)$ in $\mathcal{D}^{1,p}(\mathbb{R}^N)$. Notice that $\mathcal{D}^{1,p}_{0}(\Omega) = W_{0}^{1,p}(\Omega)$ when $\Omega$ is bounded.

The corresponding energy functional of~\eqref{main} is
\[
E(u) = \int_{\Omega} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x.
\]
Taking into account~\eqref{HSI}, we see that $E$ is well-defined on the space
\[
\mathcal{W}(\Omega)=
\begin{cases}
\mathcal{D}^{1,p}_{0}(\Omega)											&\text{if $-p<\alpha\le 0$},\\
\mathcal{D}^{1,p}_{0}(\Omega) \cap L^{p^*_\alpha}(\Omega, \abs{x}^\alpha) &\text{if $\alpha > 0$},
\end{cases}
\]
where the space
\[
L^{p^*_\alpha}(\Omega, \abs{x}^\alpha) = \braces[\Bigg]{u\colon\Omega\rightarrow\mathbb{R} \ \text{is measurable such that} \ \int_{\mathbb{R}^N} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x<+\infty}
\]
is endowed with the norm $\norm{u}_{L^{p^*_\alpha}(\omega, \abs{x}^\alpha)}=\parens[\big]{\int_{\mathbb{R}^N} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x}^\frac{1}{p^*_\alpha}$.

In the scope of this paper, we are only interested in finite energy solutions to~\eqref{main}, which are defined as critical points of $E$ in $\mathcal{W}(\Omega)$. More precisely, we call $u\in\mathcal{W}(\Omega)$ a weak solution to~\eqref{main} if
\begin{equation}\label{weakform}
\int_{\Omega} \abs{\nabla u}^{p-2} \nabla u \cdot \nabla \varphi \,\dd x = \int_{\Omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha-2}u \varphi \,\dd x \quad \text{for all $\varphi\in C^1_c(\Omega)$}.
\end{equation}

If $\Omega$ is radially symmetric, we denote by $\mathcal{D}_{{0,\rad}}^{1,p}(\Omega)$ (resp. $W_{{0,\rad}}^{1,p}(\Omega)$) the space of radial functions in $\mathcal{D}_{0}^{1,p}(\Omega)$ (resp. $W_{0}^{1,p}(\Omega)$). Then $\mathcal{D}_{{0,\rad}}^{1,p}(\Omega) \subset \mathcal{W}(\Omega)$.

By exploiting Moser's iteration technique and the standard elliptic regularity (see~\cite{MR4194414} and also~\cite{MR727034,MR709038,MR969499}), for solution $u$ to~\eqref{main}, we have
\[
\begin{aligned}
u \in C^1(\overline\Omega) \cap L^\infty(\Omega) \quad 				&\text{if $\alpha\ge0$},\\
u \in C^1(\overline\Omega\setminus\{0\}) \cap L^\infty(\Omega) \quad 	&\text{if $-p<\alpha<0$}.
\end{aligned}
\]
When $\Omega=\mathbb{R}^N$, there is no boundary condition and hence problem~\eqref{main} reads as
\begin{equation}\label{entire}
-\Delta_p u = \abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u \quad \text{in $\mathbb{R}^N$}.
\end{equation}
Positive $\mathcal{D}^{1,p}(\mathbb{R}^N)$ solutions to~\eqref{entire} with $-p<\alpha<0$ have been classified recently in~\cite{MR4611678} (see also~\cite{MR4343877}). More precisely, every positive $\mathcal{D}^{1,p}(\mathbb{R}^N)$ solution to~\eqref{entire} with $-p<\alpha<0$ must be of the form
\begin{equation}\label{sol}
U_\lambda(x) = \parens*{\frac{(N+\alpha)^\frac{1}{p} \parens[\bigg]{\frac{N-p}{p-1}}^\frac{p-1}{p} \lambda}{\lambda^p + \abs{x}^\frac{p+\alpha}{p-1}}}^\frac{N-p}{p+\alpha}
\end{equation}
for some $\lambda>0$. Moreover, in the case $\alpha>0$, all radial positive solutions to~\eqref{entire} are given by formula~\eqref{sol} (see~\cite{MR1695021}).

By Pohozaev's identity~\cite{MR1695021,MR1009077}, it is also well-known that problem~\eqref{main} has no positive solutions if $\Omega$ is a bounded star-shaped domain about the origin. However, \eqref{main} may have positive solutions if $\Omega$ is not star-shaped. Indeed, it was proved in~\cite{MR3250483} that problem~\eqref{main} with $\alpha=0$ admits a positive solution on a bounded non-contractible domain with a small hole. This result extends the classical one of Coron~\cite{MR762722}.

In this paper, we study possibly sign-changing solutions to problem~\eqref{main} and~\eqref{entire}. Since the class of sign-changing solutions may have a very complex structure, a phenomenon that has been seen in the case $\alpha=0$ (see~\cite{MR3527622,MR2825341,MR863646} for $p=2$ and~\cite{MR3788628} for $p\ne2$), there is little hope to classify all possibly sign-changing solutions to these critical problems in general domains. Nevertheless, we have the following Liouville type theorem for quasilinear Hardy--H\'enon equations on the halfspace $\mathbb{R}_{+}^N = \braces[\big]{ x \in \mathbb{R}^N : x_N > 0 }$.

\begin{theo}\label{th:halfspace}
Assume\/ $\Omega=\mathbb{R}_{+}^N$ and $\alpha>0$. Let $u \in \mathcal{W}(\mathbb{R}_{+}^N)$ be a possibly sign-changing weak solution of problem~\eqref{main}. Then $u\equiv0$.

The same conclusion holds for the case $-p<\alpha<0$ if we further assume that $\abs[\big]{\nabla u(x)} = o\parens[\big]{\abs{x}^\frac{1-N}{p}}$ as $\abs{x}\to0$.
\end{theo}

To prove Theorem~\eqref{th:halfspace} we derive a local Pohozaev type identity by following the ideas in~\cite{MR2644751,MR3989957,MR688279}. However, we use a regular form of $\partial_1 u$ instead of $\partial_N u$ as a test function. This new choice of test function allows us to exploit the assumption $\alpha\ne0$ and get the conclusion. Notice that the case $\alpha=0$, mentioned in~\cite{MR3989957}, is still an open question. To the best of our knowledge, the best Liouville type theorem for possibly sign-changing solutions in the case $\alpha=0$ is~\cite[Theorem~1.5]{MR3989957}, where the authors proved the result under the additional assumption that the Morse index of the solution is not greater than 1.

Next, we focus on radial solutions and try to derive some classification results. Problem of this type has been studied by Ni in~\cite{MR674869} in the case $p=2$. The following results extend the corresponding ones in~\cite{MR3989957} for $\alpha=0$ to the case $\alpha>-p$.

\begin{theo}\label{th:radial1}
Assume\/ $\Omega$ is a ball centered at the origin. Let $u \in W_{{0,\rad}}^{1,p}(\Omega)$ be a possibly sign-changing radial weak solution to problem~\eqref{main}. Then $u \equiv 0$.
\end{theo}

\begin{theo}\label{th:radial2}
Let $u \in \mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)\setminus\{0\}$ be a possibly sign-changing radial weak solution to problem~\eqref{entire}. Then, either $u \equiv U_\lambda$ or $u \equiv -U_\lambda$ for some $\lambda>0$.
\end{theo}

Theorem~\ref{th:radial2} allows us to derive the full description of the Palais--Smale sequences to the functional
\[
J (u) = \int_{\Omega} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} + a(x) \frac{\abs{u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x
\]
in the radial setting (see Section~\ref{sect6} for more details).

In a non-radial setting, we prove some classification results via the Morse index. For $p\ge2$, we can compute the second variation of $E$ as
\[
E''(u)[v,v] = \int_{\Omega} \bracks[\big]{\abs{\nabla u}^{p-2} \abs{\nabla v}^2 + (p-2) \abs{\nabla u}^{p-4} (\nabla u\cdot \nabla v)^2 - (p^*_\alpha - 1) \abs{x}^\alpha \abs{u}^{p^*_\alpha - 2} \abs{v}^2}\dd x
\]
for all $v \in C^1_c(\Omega)$. As in~\cite{MR2322150,MR2542716}, we say that $u$ has \emph{Morse index} $i(u)$ if $i(u)$ is the maximal dimension of the subspaces $V$ of $C^1_c(\Omega)$ such that
\[
E''(u)[v,v] < 0 \quad \text{for all $v \in V\setminus\{0\}$}.
\]

We have the following classification result for solutions with Morse index up to 1.

\begin{theo}\label{th:Morse1}
Assume $2 \le p < N$ and $-p<a<0$. Let $u \in \mathcal{D}^{1,p}(\mathbb{R}^{N}) \setminus \{0\}$ be a weak solution of problem~\eqref{entire} such that $i(u) \leq 1$. Then, either $u \equiv U_\lambda$ or $u \equiv -U_\lambda$ for some $\lambda>0$.
\end{theo}

We do not know, however, if solutions $U_{\lambda}$ have an index exactly equal to 1. This is still an open question even in the case $\alpha=0$.

For bounded domains, we have the following Liouville type result.

\begin{theo}\label{th:Morse2}
Assume $2 \le p < N$ and $\Omega$ is a smooth bounded domain of\/ $\mathbb{R}^N$, star-shaped about the origin, namely $x \cdot \nu \geq 0$ on $\partial \Omega$, where $\nu$ is the exterior normal unit vector. Let $u \in \mathcal{W}(\Omega)$ be a weak solution to problem~\eqref{main} such that $i(u) \leq 1$. Then $u \equiv 0$.
\end{theo}

The remainder of this paper is organized as follows. In Section~\ref{sect2} we prove Liouville type theorems on half space. In Section~\ref{sect3} we prove some claims on the nodal regions of solutions. In Section~\ref{sect4} and Section~\ref{sect5} we classify solutions via radial symmetry and Morse index, respectively. Section~\ref{sect6} is devoted to the descriptions for the radial Palais--Smale sequences.

\section{Liouville type theorem on a halfspace}\label{sect2}

In this section, we prove Theorem~\ref{th:halfspace}. To this end, we always assume that $u \in \mathcal{W}(\mathbb{R}_{+}^{N})$ is a weak solution of problem~\eqref{main}. Since $u \in C^1(\overline{\mathbb{R}^N_+}) \cap L^\infty(\mathbb{R}^N_+)$ if $\alpha\ge0$ and $u \in C^1(\overline{\mathbb{R}^N_+}\setminus\{0\}) \cap L^\infty(\mathbb{R}^N_+)$ if $-p<\alpha<0$, we may extend $u$ by odd reflection, i.e., $u(x',x_N)=-u(x',-x_N)$ for $x_N<0$. Therefore, in this section we can assume that $u \in \mathcal{W}(\mathbb{R}^{N})$ is a solution to~\eqref{entire} that satisfies the condition $u=0$ on $\partial\mathbb{R}_{+}^{N}$.

Hereafter we set $B_R = \braces[\big]{ x \in \mathbb{R}^N: \abs{x} < R }$ and denote by $\nu(\,\cdot\,)$ the exterior unit normal to $\partial (B_R \cap \mathbb{R}^N_+)$ whose the first component is $\nu_1 (\,\cdot\,)$. The partial derivative in $x_1$ will be denoted by~$\partial_1$.

\begin{proof}[Proof of Theorem~\ref{th:halfspace}]
Following~\cite[p.~833]{MR709038}, we take $R>\delta>0$ and let $u_{\varepsilon}$ be the unique solution to the nondegenerate boundary value problem
\begin{equation}\label{regularized}
\begin{cases}
- {\divergence} \parens[\bigg]{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} } = \abs{x}^\alpha \abs{u}^{p^*_\alpha -2}u 		& \text{in $B_{2R}\setminus B_{\delta/2}$}, \\
u_{\varepsilon} = u 					& \text{on $\partial (B_{2R}\setminus B_{\delta/2})$}.
\end{cases}
\end{equation}
Then $u_{\varepsilon} \in C^2(\overline{B_R}\setminus B_\delta)$ and uniformly bounded for $\varepsilon \in (0, 1]$ in $C^{1,\gamma}(\overline{B_R}\setminus B_\delta)$ for some $0<\gamma<1$. The Ascoli--ArzelÃ  theorem implies the existence of a suitable sequence $\varepsilon \to 0^+$ such that $u_{\varepsilon} \to u$ and $\nabla u_{\varepsilon} \to \nabla u$ uniformly on $\overline{B_R}\setminus B_\delta$.

Let us define
\[
v_{\varepsilon} = \partial_1 u_{\varepsilon} \parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon}.
\]
Since
\[
\begin{split}
{\divergence} (v_{\varepsilon}) &= \partial_1 u_{\varepsilon} \,\divergence\parens[\bigg]{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} } + \parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} \cdot \nabla \partial_1 u_{\varepsilon}\\
&= \partial_1 u_{\varepsilon} \,\divergence\parens[\bigg]{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} } + \partial_1 \parens[\Bigg]{\frac{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p}{2}}}{p}},
\end{split}
\]
by the divergence theorem, we have
\begin{multline}\label{t1}
\int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \partial_1 u_{\varepsilon} \,\divergence\parens[\bigg]{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} } \dd x \\
= \int_{\partial\parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} v_{\varepsilon} \cdot \nu(\sigma) \,\dd \sigma - \int_{\partial \parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p}{2}}}{p} \nu_1(\sigma) \,\dd \sigma.
\end{multline}
On the other hand,
\begin{equation}\label{t2}
\begin{split}
& \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \partial_1 u \,\abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u \,\dd x \\
& \hspace{3cm} = \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \partial_1 \parens[\Bigg]{\frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} } \dd x - \frac{\alpha}{p^*_\alpha} \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x \\
& \hspace{3cm} = \int_{\partial\parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} \nu_1(\sigma) \,\dd \sigma - \frac{\alpha}{p^*_\alpha} \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x.
\end{split}
\end{equation}
From~\eqref{regularized}, we deduce that
\begin{equation}\label{t3}
\int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \partial_1 u_{\varepsilon} \,\divergence\parens[\bigg]{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p-2}{2}} \nabla u_{\varepsilon} } \dd x = - \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \partial_1 u \,\abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u \,\dd x + o(\varepsilon).
\end{equation}
Substituting~\eqref{t1} and~\eqref{t2} into~\eqref{t3}, we get
\begin{multline*}
\int_{\partial \parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} v_{\varepsilon} \cdot \nu(\sigma) \,\dd \sigma - \int_{\partial \parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{\parens[\big]{\varepsilon + \abs{\nabla u_{\varepsilon}}^2 }^{\frac{p}{2}}}{p} \nu_1(\sigma) \,\dd \sigma\\
= -\int_{\partial\parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} \nu_1(\sigma) \,\dd \sigma + \frac{\alpha}{p^*_\alpha} \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x + o(\varepsilon).
\end{multline*}
Let $\varepsilon \to 0^+$, we derive
\begin{multline*}
\int_{\partial \parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \partial_1 u \abs{\nabla u}^{p-2} \nabla u \cdot \nu(\sigma) \,\dd \sigma - \int_{\partial \parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{ \abs{\nabla u}^p} {p} \nu_1(\sigma) \,\dd \sigma \\
= -\int_{\partial\parens*{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+}} \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} \nu_1(\sigma) \,\dd \sigma + \frac{\alpha}{p^*_\alpha} \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x.
\end{multline*}
This implies
\begin{equation}\label{s2}
\begin{split}
& \frac{\alpha}{p^*_\alpha} \int_{(B_R\setminus B_\delta) \cap \mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x\\
& \hspace{2cm} = \int_{\mathbb{R}^N_+ \cap \partial (B_R\setminus B_\delta)} \bracks[\Bigg]{\partial_1 u \,\abs{\nabla u}^{p-2} \nabla u \cdot \nu(\sigma) - \frac{\abs{\nabla u}^{p}}{p} \nu_1(\sigma) + \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} \nu_1(\sigma) } \dd \sigma\\
& \hspace{2cm} \le C \int_{\mathbb{R}^N_+ \cap \partial B_\delta} \parens[\big]{\abs{\nabla u}^p + \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd \sigma + C \int_{\mathbb{R}^N_+ \cap \partial B_R} \parens[\big]{\abs{\nabla u}^p + \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd \sigma
\end{split}
\end{equation}
for some constant $C>0$.

If $\alpha>0$, then $u$ and $\abs{\nabla u}$ are bounded, hence
\begin{equation}\label{s3}
\int_{\mathbb{R}^N_+ \cap \partial B_\delta} \parens[\big]{\abs{\nabla u}^p + \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd \sigma \to 0 \quad \text{as $\delta\rightarrow 0$}.
\end{equation}

If $-p<\alpha<0$, by exploiting the assumption $\abs[\big]{\nabla u(x)} = o\parens[\big]{\abs{x}^\frac{1-N}{p}}$ as $\abs{x}\to0$, we derive $\abs[\big]{u(x)} = o\parens[\big]{\abs{x}^\frac{1-N+p}{p}}$ and
\[
\int_{\mathbb{R}^N_+ \cap \partial B_\delta} \parens[\big]{\abs{\nabla u}^p + \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd \sigma \le C\parens[\bigg]{\delta^{N-1}o\parens{\delta^{1-N}} + \delta^{N-1+\alpha}o\parens[\bigg]{s^\frac{(1-N+p)(N+\alpha)}{N-p}}}.
\]
Since
\[
N-1+\alpha + \frac{(1-N+p)(N+\alpha)}{N-p} = \frac{\alpha+p}{N-p} > 0,
\]
we deduce that~\eqref{s3} also holds in this case.

On the other hand, since $\abs{\nabla u} \in L^p(\mathbb{R}^N_+)$ and $u \in L^{p^*_\alpha}(\mathbb{R}^N_+, \abs{x}^\alpha)$, there exists a sequence of radii $R_k \to \infty$ such that
\begin{equation}\label{s4}
R_k \int_{\mathbb{R}^N_+ \cap \partial B_{R_k}} \parens[\big]{\abs{\nabla u}^p + \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd \sigma \to 0 \quad \text{as $k\rightarrow\infty$}.
\end{equation}
Hence letting $\delta\rightarrow0^+$ and $R=R_k \rightarrow \infty$ in~\eqref{s2} and using~\eqref{s3} and~\eqref{s4}, we deduce
\[
\frac{\alpha}{p^*_\alpha} \int_{\mathbb{R}^N_+} \abs{x}^{\alpha-2} x_1 \abs{u}^{p^*_\alpha} \dd x = 0.
\]
Since $\alpha\ne0$, this implies $u\equiv0$.
\end{proof}

\section{On nodal regions of solutions}\label{sect3}

We call $\omega$ a nodal domain of a continuous function $u\colon\Omega \to \mathbb{R}$ if $\omega$ is a connected component of the set $\braces[\big]{x\in\Omega : u(x) \ne 0}$. In order to use $\restr{u}{\omega}$ as a test function in~\eqref{weakform}, we need the following approximation result.

\begin{lemm}\label{lemma9}
Assume $u \in C_{{\loc}}^{0,1}(\mathbb{R}^N\setminus\{0\}) \cap \mathcal{W}(\mathbb{R}^N)$. Let $\omega$ be a nodal domain of $u$, and $\restr{u}{\omega}$ its restriction to $\omega$. Then there exists a sequence $\{ u_n \}_n \subset C_c^{0,1}(\omega) $ such that:
\begin{enumerate}\romanenumi
\item $u_n \rightarrow \restr{u}{\omega}$ in $L^{p^*_\alpha}(\omega, \abs{x}^\alpha)$ and everywhere in $\omega$;
\item $\nabla u_n \rightarrow \nabla \restr{u}{\omega}$ in $L^{p}(\omega;\mathbb{R}^N)$ and almost everywhere in $\omega$.
\end{enumerate}
\end{lemm}

The proof of Lemma~\ref{lemma9} is similar to that of~\cite[Lemma~3.1]{MR3989957} and hence will be omitted. Motivated by~\cite{MR3989957}, the main result of this section is the following claim on nodal regions of solutions.

\begin{theo}\label{th:nodal}
Let $u \in \mathcal{W}(\Omega)$ be a nontrivial solution to~\eqref{main}. Then:
\begin{enumerate}\romanenumi
\item \label{th:nodal_i} For every nodal domain $\omega$ of $u$ it holds that
\[
\int_{\omega} \abs{\nabla u}^p \dd x = \int_{\omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x.
\]
\item \label{th:nodal_ii} Let $\mathcal{N}_u$ be the set of nodal domains of $u$ and $\abs{\mathcal{N}_u}$ its cardinality. If $-p<\alpha<0$, then
\[
\abs{\mathcal{N}_u} \leq S_\alpha^{-\frac{N+\alpha}{p+\alpha}} \int_{\Omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x.
\]
In particular, $u$ has at most a finite number of nodal domains. The same conclusions hold for nontrivial solutions $u \in \mathcal{D}^{1,p}_{{0,\rad}}(\Omega)$ to~\eqref{main} in the case that $\alpha>0$ and $\Omega$ is radially symmetric.
\end{enumerate}
\end{theo}

\begin{proof}[Proof of Theorem~\ref{th:nodal}]
Since $u \in C^1(\overline{\Omega}\setminus\{0\})$, we may extend $u$ by zero outside $\Omega$ to obtain $u \in C_{{\loc}}^{0,1}(\mathbb{R}^N\setminus\{0\}) \cap \mathcal{W}(\Omega)$.

Now we take a sequence $(u_n)$ given by Lemma~\ref{lemma9} and extend $u_n$ by zero outside $\omega$. By a standard density argument for every $n \in \mathbb{N}$ we can use $u_n$ as a test function in~\eqref{weakform} to deduce
\[
\int_{\omega} \abs{\nabla u}^{p-2} \nabla u \cdot \nabla u_n \,\dd x = \int_{\omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u u_n \,\dd x.
\]
Letting $n \rightarrow \infty$ and using Lemma~\eqref{lemma9}, we obtain~\eqref{th:nodal_i}.

Now we prove~\eqref{th:nodal_ii}. Using inequality~\eqref{HSI} with $v = \restr{u}{\omega}$ extended by zero outside $\omega$ from~\eqref{th:nodal_i} we have
\[
\int_{\omega} \abs{\nabla u}^p \dd x = \int_{\mathbb{R}^N} \abs{\nabla v}^p \dd x \ge S_\alpha \parens[\Bigg]{\int_{\omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x}^\frac{p}{p^*_\alpha} = S_\alpha \parens[\Bigg]{\int_{\omega} \abs{\nabla u}^{p} \dd x}^\frac{p}{p^*_\alpha}.
\]
Hence
\[
\int_{\omega} \abs{\nabla u}^p \dd x \ge S_\alpha^\frac{N+\alpha}{p+\alpha}.
\]
It follows that
\[
\int_{\Omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \dd x = \int_{\Omega} \abs{\nabla u}^p \dd x = \sum_{\omega \in \mathcal{N}_u} \int_{\omega} \abs{\nabla u}^p \dd x \ge \sum_{\omega \in \mathcal{N}_u} S_\alpha^\frac{N+\alpha}{p+\alpha} = \abs{\mathcal{N}_u} S_\alpha^\frac{N+\alpha}{p+\alpha}.
\]
This proves~\eqref{th:nodal_ii}.
\end{proof}

\section{Classification of radial solutions}\label{sect4}

We prove Theorems~\ref{th:radial1} and~\ref{th:radial2} in this section. First, we recall that if $u$ is a solution to
\[
\begin{cases}
-\Delta_p u = \abs{x}^\alpha \abs{u}^{q-2}u 	& \text{in $\Omega$},\\
u=0											& \text{on $\partial\Omega$},
\end{cases}
\]
where $1<q\le p^*_\alpha$, then it is standard to derive the Pohozaev's identity
\[
\frac{p-1}{p} \int_{\partial\Omega} \abs{\nabla u}^p x\cdot\nu(\sigma) \,\dd \sigma = \parens[\Bigg]{\frac{N+\alpha}{q} - \frac{N-p}{p}} \int_{\Omega} \abs{x}^\alpha \abs{u}^q \dd x,
\]
see~\cite[Theorem~2.1]{MR1695021} for instance. In particular, if $u$ is a solution to~\eqref{main}, then
\begin{equation}\label{pohozaev}
\int_{\partial\Omega} \abs{\nabla u}^p x\cdot\nu(\sigma) \,\dd \sigma = 0.
\end{equation}

\begin{proof}[Proof of Theorem~\ref{th:radial1}]
Suppose $u \not\equiv 0$. We recall that $u \in C^1(\overline\Omega\setminus\{0\}) \cap L^\infty(\Omega)$.

If $u$ does not change sign, then by the strong maximum principle we may assume $u>0$ in $\Omega\setminus\{0\}$. Since $\Omega$ is a ball, identity~\eqref{pohozaev} implies $\abs{\nabla u} = 0$ on $\partial\Omega$. However, this contradicts Hopf's boundary point lemma.

Therefore, $u$ must change the sign. Since $u$ is radial, the nodal regions of $u$ are spherically symmetric. By Theorem~\ref{th:nodal}, the number of such regions is finite. We can take an annular nodal region $A = \braces[\big]{x \in \Omega: R_1 < \abs{x} < R_2}$ with $R_2 > R_1 >0$ and assume $u>0$ in $A$. Then $u$ solves
\begin{equation}
\begin{cases}
-\Delta_p u = \abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u		& \text{in $B_{R_2}$}, \\
u > 0 											& \text{in $A$}, \\
u = 0 											& \text{on $\partial A$}.
\end{cases}
\end{equation}
By Pohozaev's identity~\eqref{pohozaev} again, we have $\abs{\nabla u} = 0$ on $\partial B_{R_2}$. This again contradicts Hopf's boundary point lemma. The proof is concluded.
\end{proof}

\begin{proof}[Proof of Theorem~\ref{th:radial2}]
If $u$ does not change the sign, then $\abs{u}>0$ by the strong maximum principle. Hence~\cite[Theorem~3.1]{MR1695021} can be applied to yield $u \equiv U_\lambda$ or $u \equiv -U_\lambda$ for some $\lambda>0$.

It remains to consider the case that $u$ changes sign. By radial symmetry and Theorem~\ref{th:nodal}, the nodal regions are spherically symmetric and their number is finite. Hence, by replacing $u$ by $-u$ if necessary, we find $R > 0$ large enough such that
\begin{align*}
& u = 0 \quad \text{on } \partial B_R, \\
& u > 0 \quad \text{in } \mathbb{R}^N \setminus \overline{B_R},
\end{align*}
and so, by Theorem~\ref{th:radial1},
\begin{align*}
& u = 0 \quad \text{in } \overline{B_R}, \\
& u > 0 \quad \text{in } \mathbb{R}^N \setminus \overline{B_R}.
\end{align*}

On the other hand, by continuity of $\nabla u$, we have $\abs{\nabla u} = 0$ on $\partial B_R$. However, this is in contradiction with Hopf's boundary point lemma on $\mathbb{R}^N \setminus \overline{B_R}$. This finishes the proof.
\end{proof}

\section{Classification of solutions with Morse index at most 1}\label{sect5}

Motivated by~\cite{MR3989957}, we show that the number of nodal regions of a solution is bounded by its own Morse index.

\begin{theo}\label{theorem2}
Assume\/ $2 \le p < N$ and\/ $\Omega$ is a smooth domain of\/ $\mathbb{R}^N$. Let $u \in \mathcal{W}(\Omega)$ be a weak solution to problem~\eqref{main}. Then
\[
\abs{\mathcal{N}_u} \leq i(u),
\]
where $i(u)$ is the Morse index of $u$.
\end{theo}

\begin{proof}
Consider a nodal region $\omega$. Let $(u_n)$ be given by Lemma~\ref{lemma9} and extend $u_n$ by zero outside~$\omega$. By Lemma~\ref{lemma9} we have
\[
\begin{split}
E''(u)[u_n, u_n] &= \int_{\omega} \bracks[\big]{\abs{\nabla u}^{p-2} \abs{\nabla u_n}^2 + (p-2) \abs{\nabla u}^{p-4} (\nabla u\cdot \nabla u_n)^2 - (p^*_\alpha - 1) \abs{x}^\alpha \abs{u}^{p^*_\alpha - 2} \abs{u_n}^2} \dd x\\
& \quad \to \int_{\omega} \bracks[\big]{(p-1) \abs{\nabla u}^p - (p^*_\alpha - 1) \abs{x}^\alpha \abs{u}^{p^*_\alpha}} \dd x \quad \text{as } n\rightarrow\infty.
\end{split}
\]
This fact and Theorem~\ref{th:nodal} (i) yield
\[
E''(u)[u_n, u_n] \to (p- p^*_\alpha) \int_{\omega} \abs{x}^\alpha \abs{u}^{p^*_\alpha} \,\dd x < 0.
\]
Therefore, for every nodal region $\omega$ there exists $v \in C_c^{0,1}(\omega)$ (and by density in $C_c^1(\omega)$) such that
\[
E''(u)[v, v] < 0.
\]
By the definition of Morse index, this implies $\abs{\mathcal{N}_u} \leq i(u)$.
\end{proof}

Now we can prove our classification results for solutions $u$ with Morse index $i(u) \le 1$.

\begin{proof}[Proof of Theorem~\ref{th:Morse1}]
By Theorem~\ref{theorem2}, $u$ has exactly one nodal region, which is denoted by $A$.

If $A$ is a proper subset and $A \ne \mathbb{R}^N \setminus\{0\}$, we can assume, up to consider $-u$ instead of $u$, that $A = \{u > 0\}$. Since $u \in C^1(\mathbb{R}^N\setminus\{0\})$, we have $\abs{\nabla u} = 0$ on $\partial A\setminus\{0\}$. We can pick an interior point $p \in A$ and a radius $R>0$ such that the ball $B_R(p) \subset A$ touches $\partial A$ at some point $p'\ne0$. Now we can apply Hopf's boundary point lemma at boundary point $p'$ for equation on $B_R(p)$ to get $\abs[\big]{\nabla u(p')} \neq 0$. This is a contradiction.

Therefore, either $A = \mathbb{R}^N$ or $A = \mathbb{R}^N \setminus\{0\}$. Then the conclusion follows by a recent classification result in~\cite{MR4611678,MR4343877}. This concludes the proof.
\end{proof}

\begin{proof}[Proof of Theorem~\ref{th:Morse2}]
Since $\Omega$ is star-shaped, we may use~\eqref{pohozaev} to deduce that the normal derivative $\partial_\nu u$ vanishes at some point $x_0 \in \partial \Omega$. On the other hand, Theorem~\ref{theorem2} implies that $u$ has at most one nodal region. If $u$ were nontrivial, we get a contradiction with Hopf's boundary point lemma. Hence it must happen that $u \equiv 0$.
\end{proof}

\section{Descriptions for the Palais--Smale sequences in the radial setting} \label{sect6}

In this section, we consider the problem
\begin{equation}\label{main_a}
\begin{cases}
-\Delta_p u + a(x) \abs{u}^{p-2} u = \abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u &\text{in } \Omega,\\
u=0 &\text{on } \partial\Omega,
\end{cases}
\end{equation}
where $-p<\alpha<0$, $\Omega$ is bounded and $a \in L^\frac{N}{p}(\Omega)$. Solutions to~\eqref{main_a} coincides with the critical points in $W_{0}^{1,p}(\Omega)$ of the functional
\[
J (u) = \int_{\Omega} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} + a(x) \frac{\abs{u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x.
\]
We also define on $\mathcal{D}^{1,p}(\mathbb{R}^N)$ the limit functional
\[
J_\infty (u) = \int_{\mathbb{R}^N} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x.
\]
We recall that $\{ u_n \}_n \subset W_{0}^{1,p}(\Omega)$ is called a Palais--Smale sequence of $J$ at energy level $c\in\mathbb{R}$ if
\[
J (u_n) \rightarrow c \quad\text{and}\quad J' (u_n) \rightarrow 0 \quad \text{in } W^{-1,p'}(\Omega).
\]
The compactness of Palais--Smale sequences is a vital condition in the theory of critical points. Using the global compactness results in~\cite{MR2746823} and an idea in~\cite{MR3989957} we can give a precise description of Palais--Smale sequences for functional $J$ on the unit ball as follows.
\begin{prop}
Assume $-p<\alpha<0$, $\Omega$ is the unit ball centered at the origin in\/ $\mathbb{R}^N$ and $a \in L_{\rad}^\frac{N}{p}(\Omega)$ such that
\[
\inf_{\substack{u \in W_{0}^{1,p}(\Omega) \\ \norm{u}=1}} \int_{\Omega} \bracks[\big]{\abs{\nabla u}^{p} + a(x) \abs{u}^p }\dd x > 0.
\]
Let $\{u_n \}_n$ be a sequence in $W_{{0,\rad}}^{1,p}(\Omega)$ such that
\[
J (u_n) \rightarrow c \quad\text{and}\quad J' (u_n) \rightarrow 0 \quad \text{in } \parens[\big]{W_{{0,\rad}}^{1,p}(\Omega)}'.
\]
Then, passing if necessary to a subsequence, there exists a possibly trivial solution $v_0 \in W_{{0,\rad}}^{1,p}(\Omega)$ to~\eqref{main_a} and $k$ sequences $\{ \lambda_n^i \}_n \subset \mathbb{R}_{+}$, $i=1,\dotsc,k$, such that $\lambda_n^i \to 0$ as $n \to \infty$ and
\begin{align}
\norm[\Bigg]{u_n - v_0 - \sum_{i=1}^{k} (\lambda_n^i)^\frac{p-N}{p} v_i \parens[\Bigg]{\frac{\cdot}{\lambda_n^i}}} \to 0,& \quad n \rightarrow \infty, \label{PS1}
\\
\norm{u_n}^p \to \sum_{i=0}^{k} \norm{v_i}^p,& \quad n \rightarrow \infty, \label{PS2}
\\
J (v_0) + \sum_{i=1}^{k} J_\infty(v_i) = c,& \label{PS3}
\end{align}
where for each $i=1,\dotsc,k$, either $v_i \equiv V_{\mu_i}$ or $v_i \equiv -V_{\mu_i}$ for some $\mu_i > 0$ and
\[
J_\infty(v_i) = \frac{p+\alpha}{(N+\alpha)p}S_\alpha^\frac{N+\alpha}{p+\alpha}.
\]
Moreover if $a \equiv 0$, then all weakly convergent subsequences of $\{ u_n\}_n$ are weakly convergent to zero in $W_{{0,\rad}}^{1,p}(\Omega)$. In particular, $v_0 \equiv 0$ and hence $J (v_0) = 0$.
\end{prop}

\begin{proof}
By~\cite[Theorem~3.1]{MR2746823}, we know that up to a subsequence, there exists a possibly trivial solution $v_0 \in W_{{0,\rad}}^{1,p}(\Omega)$ to~\eqref{main_a}, $k$ nontrivial solutions $\{ v_1,\dotsc,v_k \} \subset \mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)$ to~\eqref{entire} and $k$ sequences $\{ \lambda_n^i \}_n \subset \mathbb{R}_{+}$, such that $\lambda_n^i \to 0$ as $n \to \infty$ and~\eqref{PS1}--\eqref{PS3} is satisfied. By Theorem~\ref{th:radial2}, we have either $v_i \equiv V_{\mu_i}$ or $v_i \equiv -V_{\mu_i}$ for some $\mu_i > 0$. Hence $J_\infty(v_i)$ can be computed explicitly as in~\cite{MR463908}.

Assume now $a \equiv 0$. Then $v_0$ solves~\eqref{main} in the unit ball. By Theorem~\ref{th:radial1}, we deduce $v_0 \equiv 0$. Since $v_0$ is the weak limit of $\{ u_n\}_n$, the conclusion follows.
\end{proof}

We can state a similar result for the problem
\begin{equation}\label{main_b}
-\Delta_p u + a(x) \abs{u}^{p-2} u = \abs{x}^\alpha \abs{u}^{p^*_\alpha-2} u \quad\text{in } \mathbb{R}^N.
\end{equation}
Solutions in $\mathcal{D}^{1,p}(\mathbb{R}^N)$ to~\eqref{main_b} coincide with the critical points in $\mathcal{D}^{1,p}(\mathbb{R}^N)$ of the functional
\[
J (u) = \int_{\mathbb{R}^N} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} + a(x) \frac{\abs{u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x.
\]
As before, we also define on $\mathcal{D}^{1,p}(\mathbb{R}^N)$ the functional
\[
J_\infty (u) = \int_{\mathbb{R}^N} \parens[\Bigg]{\frac{\abs{\nabla u}^p}{p} - \frac{\abs{x}^\alpha \abs{u}^{p^*_\alpha}}{p^*_\alpha} }\dd x.
\]
We have the following result.
\begin{prop}
Assume $-p<\alpha<0$ and $a \in L_{\rad}^\frac{N}{p}(\mathbb{R}^N)$ such that
\[
\inf_{\substack{u \in \mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N) \\ \norm{u}=1}} \int_{\mathbb{R}^N} \bracks[\big]{\abs{\nabla u}^{p} + a(x) \abs{u}^p }\dd x > 0.
\]
Let $\{u_n \}_n$ be a sequence in $\mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)$ such that
\[
J (u_n) \rightarrow c \quad\text{and}\quad J' (u_n) \rightarrow 0 \quad \text{in } \parens[\big]{\mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)}'.
\]
Then, passing if necessary to a subsequence, there exists a possibly trivial solution $v_0 \in \mathcal{D}^{1,p}_{\rad}(\mathbb{R}^N)$ to~\eqref{main_b} and $k$ sequences $\{ \lambda_n^i \}_n \subset \mathbb{R}_{+}$, $i=1,\dotsc,k$, such that $\lambda_n^i \to 0$ as $n \to \infty$ and
\begin{align*}
\norm[\Bigg]{u_n - v_0 - \sum_{i=1}^{k} (\lambda_n^i)^\frac{p-N}{p} v_i \parens[\Bigg]{\frac{\cdot}{\lambda_n^i}}} \to 0,& \quad n \rightarrow \infty, \\
\norm{u_n}^p \to \sum_{i=0}^{k} \norm{v_i}^p,& \quad n \rightarrow \infty, \\
J (v_0) + \sum_{i=1}^{k} J_\infty(v_i) = c,&
\end{align*}
where for each $i=1,\dotsc,k$, either $v_i \equiv V_{\mu_i}$ or $v_i \equiv -V_{\mu_i}$ for some $\mu_i > 0$ and
\[
J_\infty(v_i) = \frac{p+\alpha}{(N+\alpha)p}S_\alpha^\frac{N+\alpha}{p+\alpha}.
\]
Moreover if $a \equiv 0$, then either $v_0$ is identically zero, or it holds that for some $\lambda > 0$ and up to the sign $v_0 \equiv V_\lambda$ and
\[
J(v_0) = \frac{p+\alpha}{(N+\alpha)p}S_\alpha^\frac{N+\alpha}{p+\alpha}.
\]
\end{prop}

\begin{proof}
It follows from~\cite[Theorem~4.1]{MR2746823} and Theorem~\ref{th:radial2}. Note that $J_\infty(v_i)$ can be computed explicitly by~\cite{MR463908}, using Theorem~\ref{th:radial2}.
\end{proof}

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography

\end{document}